Author’s response to reviews

Title: Clear lens phacoemulsification in anterior lenticous due to Alport syndrome’case reports’ Ghassem Amiraslanzadeh, MD, Assistant prof. ophthalmology, Nikookari Eye Center, Tabriz University of Medical Sciences, Tabriz, Iran Corresponding author dr.aslanzadeh@gmail.com Davoud Gharabaghi, MD, Associated prof. ophthalmology, Nikookari Eye Center, Tabriz University of Medical Sciences, Tabriz, Iran gharabaghid@gmail.com Niloofar Naderi, MD, resident of ophthalmology Nikookari Eye Center, Tabriz University of Medical Sciences, Tabriz, Iran dr_niloofarnaderi@yahoo.com

Authors:

Ghassem Amiraslanzadeh (dr.aslanzadeh@gmail.com)
Davood Gharabaghi (gharabaghid@gmail.com)
Niloofar Naderi (dr_niloofarnaderi@yahoo.com)

Version: 2 Date: 11 August 2007

Author’s response to reviews: see over
Dear editorial staff thank you for your valuable guides. This article include presentation of 2 cases anterior lenticonus due to Alport syndrome. This is a somewhat rare syndrome that emerge from a genetic defect in synthesis of collagen 4, the maine substance of basement membranes in the human body. We think that it can be usefull in educational field. Additionally the procedure that we (and some others) preformed and recommended can be a protocol of choice in this type disorders.